Author's response to reviews

Title: The foot posture index, ankle lunge test, Beighton scale and the lower limb assessment score in healthy children: a reliability study

Authors:

   Angela M Evans (angela.evans@unisa.edu.au)
   Keith Rome (krome@aut.ac.nz)
   Lauren Peet (lpeet@aut.ac.nz)

Version: 4 Date: 5 January 2012

Author's response to reviews:

Dear JFAR

Thank you for your assistance with our manuscript.
All major and minor revisions have been made as requested:
MAJOR
- the author list is amended as requested
- the affiliations are amended as requested
- the latest abstract has been uploaded in its corrected form
MINOR
- qualifications have been removed from the authors list as requested
- email addresses are formatted as requested
- typography has been checked (one space added)

Please let me know if anything further is required.
Thank you once again for your assistance,
Kind regards

Angela Evans
6/1/12